

FISCAL SUSTAINABILITY: ISSUES FOR EMERGING MARKET COUNTRIES

Alejandro Izquierdo & Ugo Panizza Working Paper No. 91 December 2003

Alejandro Izquierdo and Ugo Panizza are economists in the research department of the Inter-American Development Bank in Washington D.C. This paper was prepared for the ECES conference on "Fiscal Sustainability and Public Expenditures in Egypt," which was held in Cairo on October 19-20, 2003. The authors would like to thank George Abed, Hanaa Kheir El Din, Marcelo Oviedo, Ludvig Soderling, and the conference participants for very helpful comments and suggestions, Carlos Diaz Alvarado for his dedicated research assistance, Mona Ghander for very valuable help with data, and Ahmed Galal for his invitation to write this paper. The views expressed in this paper are the authors' and do not necessarily reflect those of the Inter-American Development Bank. The usual caveats apply.

Abstract

This paper surveys the recent literature on fiscal sustainability with particular focus on emerging market countries. It discusses the main elements that differentiate emerging market countries from industrial countries and then discusses how probabilistic models can help to evaluate fiscal sustainability in an uncertain environment. Based on this discussion, the paper uses Egypt to illustrate an application of the probabilistic model and to evaluate the impact of shocks to current account financing on sustainability.

.

•

(probabilistic models)

.

I. INTRODUCTION

The purpose of this paper is to review the recent literature on fiscal sustainability with particular reference to the problems that are specific to emerging market countries (EMs). While the original literature on fiscal sustainability mostly focused on industrial countries (see, for instance, Buiter, 1985; Blanchard, 1990), there are now a few pieces like this one that focus on fiscal sustainability in EMs.¹

The paper focuses on the role of currency and maturity mismatches, original sin, sudden stops in capital flows, debt intolerance, and concessional debt. After discussing standard models of fiscal sustainability, the paper highlights the importance of moving from deterministic models of fiscal sustainability to probabilistic models, and describes in greater detail the probabilistic model developed by Mendoza and Oviedo (2003).

The paper is organized as follows. Section 2 introduces the concept of fiscal sustainability and discusses its main definitions and the main sustainability indicators that have been proposed in the literature. Section 3 focuses on the case of EMs and discusses what makes these countries different from industrial countries. Section 4 briefly reviews the main stochastic models of fiscal sustainability. Section 5 examines the case of Egypt, and includes three different approaches to fiscal sustainability. Section 6 summarizes the study's main findings.

2. WHAT DO WE MEAN BY FISCAL SUSTAINABILITY?

The term "fiscal sustainability" is often used without having a clear definition in mind. A country's policies are defined as fiscally sustainable if they lead to a situation in which the country can satisfy its budget constraint. However, Mendoza (2003) suggests that this is an imprecise definition of sustainability. He points out that the "true" budget constraint is an accounting identity that, by definition, is always satisfied. A government, for instance, can decide to satisfy its budget constraint by not paying (via outright default) or by inflating away its debt. In this sense, any analysis of fiscal sustainability ultimately reflects a value judgment on the cost and benefits of alternative adjustment mechanisms. So, standard

¹ Work that is closely related to ours is Mendoza (2003). Other relevant papers include IMF (2002, 2003b), Chalk and Hemming (2000), and Cuddington (1996). Alba, Al Shawarby, and Iqbal (Forthcoming) focus on fiscal sustainability in Egypt.

sustainability analysis implicitly assumes that adjustments through the level and composition of tax revenues or primary expenditure are preferable to adjustments via default or inflation (Mendoza, 2003).

IMF (2002) and Croce and Juan-Ramon (2003) discuss the difference between solvency² and sustainability. According to their definition, a set of policies is unsustainable if it leads to insolvency. However, they suggest that solvency is only a necessary condition for sustainability because it could be achieved with very large and costly future adjustments. Sustainability, instead, requires achieving solvency with unchanged policies. So, we can define a policy stance as sustainable if "a borrower is expected to be able to continue servicing its debt without an unrealistically large future correction to the balance of income and expenditure" (IMF, 2002, 4).

With these considerations in mind, we define a sustainable situation as one that satisfies the following two conditions: (i) if a country can satisfy its current period budget constraint without recurring to default or excessive debt monetization; and (ii) if a country does not keep accumulating debt by knowing that a major future adjustment will be needed in order to be able to service its debt.

2.1 More Definitions

Up to this point, we made use of two terms that are key in the fiscal sustainability debate without giving proper definitions. The terms are current period budget constraint and intertemporal budget constraint.

The current period budget constraint is an expression that equates the flows of government revenues and expenditures to changes in the stock of public debt and in the monetary base. Formally:

$$(D_{t+1} - D_t) + (M_{t+1} - M_t) = iD_t + G_t - REV_t$$
(1)

 $^{^2}$ Solvency is defined as a situation in which the future paths of spending and revenue satisfy the intertemporal budget constraint.

where *D* measures the stock of public debt (measured at the beginning of the period), *M* is the monetary base, *i* is the interest rate paid by government debt, *G* is government expenditure in goods and services, and *REV* represents taxation (net of transfers) and other revenues (they could be royalties from natural resources). Equation (1) clearly shows that a given deficit can be financed either by issuing debt (bond financing) or by printing money (money financing). As excessive money financing may lead to hyperinflation, equation (1) is often written as $(D_{t+1} - D_t) = iD_t + G_t - REV_t$. It should be clear that the equation does not impose a strong constraint on governments that are able to issue debt. In fact, Wilcox (1989, 291) points out that "virtually any pattern of deficit would be sustainable if it were possible to borrow money and pay the interest by borrowing more."

The inter-temporal constraint, instead, imposes a limit on the government's ability to borrow indefinitely, by requiring net initial debt plus the present value of expected future government expenditures to be equal to (or not greater than) the present value of expected future government revenues. Formally:

$$D_{t} + \sum_{k=0}^{\infty} \frac{E_{t} (G_{t+k} + iD_{t+k})}{(1+i)^{k}} \le \sum_{k=0}^{\infty} \frac{E_{t} (REV_{t+k})}{(1+i)^{k}}$$
(2)

where E_t denotes expectation taken at time *t*, and all other variables are defined as above. Note that evaluating equation (2) requires formulating expectations on the future path of government revenues and expenditures. Furthermore, we greatly simplified the equation by assuming that the interest rate paid on government debt is constant and equal to the discount rate. Relaxing these assumptions would further complicate the analysis.³

³ Equation 2 would become $D_t + E_t \sum_{k=0}^{\infty} \frac{(G_{t+k} + i_{t+k}D_{t+k})}{\prod_{j=1}^k (1+\delta_{t+j})} \le E_t \sum_{k=0}^{\infty} \frac{(REV_{t+k})}{\prod_{j=1}^k (1+\delta_{t+j})}$ where δ is the time-

varying discount rate.

One implication of equation (2) is that in the limit (as t goes to infinity), the present value of debt in the terminal period should be zero. Formally, equation (2) requires that:

$$\lim_{\tau \to \infty} \frac{D_{t+\tau}}{\left(1+i\right)^{\tau}} = 0 \tag{3}$$

This condition is often referred to as no Ponzi game condition or NPG. Since sustainability requires that the above conditions be satisfied without a radical change in policies, sustainability can be tested by looking at whether the current fiscal stance will eventually lead to a violation of equation (3). Starting with Hamilton and Flavin (1986), a long series of papers used data from OECD countries to perform sustainability tests.⁴ As these types of tests require long time series of fiscal data (and these time series should not have large structural breaks), research on developing countries has been much more limited.

2.2 Sustainability Indicators

As formal tests of sustainability tend to be problematic and tend to be rather demanding in terms of data requirement, some analysts have developed rule of thumb indicators aimed at checking whether current policies can stabilize or reduce a given debt ratio. While these indicators have the advantage of being simple, it should be recognized that they are not based on any well-specified definition of sustainability.⁵

The starting point for deriving these indicators is the current period budget constraint of equation (1) that, after dividing all variables by GDP, can be re-written as:

$$\Delta d = (r - g)d - ps \tag{4}$$

where *d* is the debt to GDP ratio, *r* the steady state real interest rate, *g* the long-run growth rate of real GDP, and *ps* the primary surplus (defined as (REV-G)/GDP). A positive value

⁴ For a survey of these papers, see Cuddington (1996). Bohn (1995) was the first to explicitly include uncertainty in these tests.

⁵ For a description of the practical approach to sustainability followed by the IMF, see Chalk and Hemming (2000).

of (4) indicates that debt to GDP is expanding and may be interpreted as an unsustainable policy. After setting Δd equal to zero, equation 4 is often rewritten as ps = (r - g)d, and *ps* is interpreted as the primary surplus required to stabilize the debt-to-GDP ratio for a given real interest rate, growth rate of the economy and initial stock of debt.⁶ Given its simplicity, equation (4) is probably the most commonly used indicator of sustainability.⁷

Buiter (1985) suggests an alternative indicator of sustainability can be defined as:

$$SUS = ps - \left(g - r\right) \frac{W}{GDP}$$
(5)

where *W* is public sector net worth and all the other variables are defined as above. The first term on the right hand side of equation (5) is equal to the primary surplus that keeps the public sector wealth-to-GDP ratio constant. So, according to this indicator, sustainability depends on the difference between actual primary surplus and the surplus that stabilizes net government wealth (scaled by GDP). Negative values of *SUS* are taken as indication that the current fiscal stance is unsustainable. One advantage of equation (5) is that it explicitly assumes that government expenditure could increase government wealth (this is one of the points recently raised by some developing countries, which we will discuss in detail later). One of the main problems with equation (5) is that government net worth is very difficult to measure.

Blanchard (1990) defines a set of sustainability indicators that require computing the constant tax rate that satisfies $t^* = E(e + (r - g)d)$, where t^* measures taxes over GDP and *e* government expenditure over GDP.⁸ This technique can be used to compute short-

⁷ Equation (4) is the standard textbook formulation of the evolution of debt over GDP. A more precise definition for the evolution of debt in discrete time is $\Delta d = \frac{(r-g)}{1+g}d - ps$. With this second formulation, the standard sustainability equation becomes: $ps = \frac{(r-g)d}{1+g}$. For all practical purposes, the two definitions yield similar debt stabilizing primary surpluses.

⁸ Under certainty, as ps = t - g, this is equivalent to ps = (r - g)d.

⁶ All the indicators discussed in this section assume that (r - g) > 0, which is a necessary condition for dynamic efficiency. For a discussion on this condition, see Blanchard (1990).

run (where expectations are replaced with current values of e, r, and g) or, depending on the length of the period for which expectations are taken, medium and long-run indicators. Blanchard (1990) points out that t^* has an easy interpretation because it is equal to the annuity value of expected future spending and transfers plus the difference between expected real interest rate and growth rate multiplied by the current debt-to-GDP ratio. Then, if t^* is larger than the current tax rate (t), an adjustment in spending or taxation will be required and hence the fiscal policy stance would not be sustainable. The sustainability indicator ($t^* - t$) measures the size of the required adjustment in the current period.

Blanchard (1990) suggests that different values of $(t^* - t)$ will have different implications for sustainability depending on the starting level t. Countries with a low tax rate may have more room to adjust, while countries that already have high tax levels or limited ability to raise taxes (maybe because of the presence of a large informal sector, as it often happens in developing countries) may have to resort to debt monetization or outright default.⁹

There are several caveats that apply to the indicators discussed above. First, they mostly focus on stabilizing a particular debt-to-GDP ratio but they do not say anything about the optimality of this ratio. Hence, some countries may need to aim at a lower debt target, and sustainability should be defined as the policy stance needed to reach this new target.

Second, all the indicators discussed so far are sufficient (but not necessary) conditions for long-run sustainability. There are good reasons why a country may want to run a large deficit. Hence, it may be sub-optimal to prevent a country from smoothing expenditure (or conducting counter-cyclical policies) because this would lead to overshooting a fiscal ratio that corresponds to a long-run equilibrium.¹⁰

Third, these indicators require assumptions on GDP growth, interest rate, government expenditures and revenues, and implicitly assume that these variables are

⁹ However, countries with larger public sector and larger tax bases have more ability to adjust. See IDB (1995).

¹⁰ For example, see the discussion on the Growth and Stability Pact within the European Monetary Union in the *Economist* (2002).

exogenous. However, most of these variables tend to be endogenous and correlated with one another. It is clearly unrealistic to assume that changes in the primary deficit will have no effect on the interest rate and growth, or that changes in growth do not affect the primary surplus. Croce and Juan-Ramon (2003) propose a recursive fiscal sustainability indicator aimed at addressing these issues. They derive the primary surplus and discount rate that would prevail when a country reaches its target debt-to-GDP ratio and then build a fiscal sustainability indicator that lets these variables react to shocks that move the debtto-GDP ratio out of its equilibrium value. This indicator identifies the reaction function of the government, and, hence, allows the analyst to evaluate whether the fiscal stance is moving toward or away from sustainability.

Finally, most of these indicators do not take into account a host of factors that characterize the situation in most emerging market and developing countries and greatly increase uncertainty.

3. WHY ARE EMERGING MARKETS DIFFERENT?

Several features make emerging market countries different from OECD countries, for which most of the indicators discussed above have been derived. In particular, EMs often have limited capacity to raise taxes (because of a large informal sector), have volatile revenues base, are subject to large external shocks (both real and financial) that increase the volatility of GDP growth and that of debt service, and are characterized by large levels of liability dollarization (IMF, 2003c). All of these elements complicate the management of fiscal policy and greatly increase the difficulty of evaluating sustainability.

In order to clarify this statement, let us start by modifying equation (4) to include some of the elements that are common to EMs:

$$\Delta d = (\alpha r^{dl} + \beta r^{ds} + \gamma \frac{(1+\rho+r^f)(1+\varepsilon)-1}{1+\pi} + (1-\alpha-\beta-\gamma)\frac{(1+r^f)(1+\varepsilon)-1}{1+\pi} - g)d - ps \quad (6)$$

where α is the share of debt denominated in local currency at a fixed (long-term) interest rate, and r^{dl} is the corresponding real interest rate. β is the share of debt denominated in local currency at a floating (short-term) interest rate and r^{ds} is the corresponding real interest rate. γ is the share of debt denominated in foreign currency, ε is nominal

depreciation, π is inflation, r^{f} is the international interest rate, and ρ is country risk. (1- $\alpha - \beta - \gamma$) is official debt contracted with multilateral or bilateral institutions.¹¹ Contrary to OECD countries, in the typical EM, β and γ tend to be high and α tends to be small. Therefore, EMs will tend to have a large share of their debt in either domestic currency at a floating rate or in foreign currency at a fixed rate (some EMs have a substantial share of concessional and official debt, but this tends to be the case for poorer developing countries).¹²

It is now easy to see how the characteristics of EMs complicate the sustainability exercises discussed in the previous section.

Real external shocks (like a terms of trade shock) tend to be larger in EMs. This affects the volatility of GDP growth and hence makes *g* difficult to estimate. There is also some evidence that the effect of external shocks is amplified by the presence of sudden stops in capital flows (Galindo and Izquierdo, 2003).

A weak fiscal position is another key characteristic of EMs and developing countries. Fiscal policies and budget institutions in EMs are often not credible and this prevents them from managing counter-cyclical policies by making credible announcements to reduce public expenditure or cut taxes in good times (IMF, 2003c).

Liability dollarization and original sin refer to the fact that in most EMs there are limited opportunities to borrow long-term in the country's own currency (hence, EMs have high values of β and γ).¹³ Because of pervasive liability dollarization, EMs tend to suffer from "fear of floating" (Calvo and Reinhart, 2002; Hausmann, Panizza and Stein, 2001) and hence tend to overstabilize the exchange rate, even if, formally, the exchange rate regime is announced as a flexible one. However, stabilizing the exchange rate requires

¹¹ We assume that this debt is contracted at the international interest rate. In some cases the actual rate will be higher (when the debt is not concessional) and in others, lower. However, this does not change our analysis as long as the interest rate applied to this type of debt has limited volatility.

¹² Edwards (2002) and IMF (2003a) focus on debt sustainability in low-income countries.

¹³ Eichengreen, Hausmann and Panizza (2003a) point out that this is especially true in the case of external debt. They define original sin as the inability of a country to borrow abroad in its own currency. In earlier work, Eichengreen and Hausmann (1999) used the term original sin to refer to both the inability of some countries to borrow abroad in their own currencies and their inability to borrow at home at long maturities. For a discussion of the determinants of original sin, see Hausmann and Panizza (2003).

large adjustments in the domestic interest rate, which amplifies uncertainty about the costs linked to servicing debt expressed in domestic currency at a floating rate. At the same time, episodes of financial contagion and the possibility of self-fulfilling fiscal crises affect country risk and increase the volatility of the cost of servicing foreign currency debt. Barnhill and Kopits (2003) point out that as the budget constraint of EMs is particularly difficult to observe, investor sentiments tend to be particularly volatile. These shocks may lead to multiple equilibria: a country that under a tranquil condition may have a perfectly sustainable policy stance may suddenly jump to an unsustainable situation because fear of default leads international investors to ask for larger risk premia. Barnhill and Kopits (2003) study the case of Ecuador and show that the volatility of sovereign spread is a major source of fiscal vulnerability (more important than terms of trade shocks). In this sense, there are instances in which the behavior of creditors is the ultimate determinant of sustainability.

Sudden stops in capital flows could also be a key determinant of sustainability. For instance, Calvo, Izquierdo and Mejía (2003), argue that a loss of access to credit markets need not be the result of over-indebtedness *in the context of a good equilibrium*, but rather the result of an economy having fallen into a bad equilibrium *triggered by a sudden stop in capital flows*. This inverse fiscal view finds support in the fact that sudden stop episodes tend to occur around the same time, and for countries exhibiting a variety of fiscal situations. Sudden stops in capital flows force abrupt adjustments of the current account deficit that may require a large adjustment (depreciation) of the real exchange rate (Calvo, Izquierdo and Talvi, 2002). This adjustment may have large valuation effects and multiply the cost of servicing foreign currency debt because of excessive liability dollarization, thus pushing a country over the edge of unsustainability.¹⁴ There is also evidence that apart from amplifying the effect of sudden stops, liability dollarization may itself be a determinant of the probability of having a sudden stop (Calvo, Izquierdo and Mejía, 2003).

¹⁴ A liquidity crisis arises when an otherwise solvent debtor does not have enough liquid assets to meet or roll-over its maturing liabilities. IMF (2002) correctly points out that liquidity may not be an issue for low income countries that, while having high debt ratio, do not need to borrow from the private capital market (because most of their external liabilities are with official creditors). However, it is a serious issue for even moderately indebted EMs.

Debt is riskier in EMs, leading to a situation in which relatively low (for industrial country standards) debt-to-GDP ratios result in very poor credit ratings. Switzerland and Costa Rica have similar debt-to-GDP ratios and so do the United States and Turkey, or Italy, Japan, and Belgium, and Jordan, Pakistan and Jamaica. However, these EMs have very different credit ratings than those of developed countries. Switzerland, the United States, Italy, Japan and Belgium have a rating of at least AA, while the EMs listed above are well below investment grade. This de-linkage between credit ratings and debt ratios has been recognized by several authors who attribute it to different causes. Hausmann (2003), and Eichengreen, Hausmann and Panizza (2003b) emphasize the role of original sin and suggest that foreign currency debt makes the cost of servicing the debt dependent on the real exchange rate, which is uncertain and pro-cyclical. As these elements increase the probability of being in a state of the world in which payment becomes very difficult, they will lead to lower credit ratings (Hausmann, 2003, discusses a formal model). Reinhart, Rogoff and Savastano (2003), instead, introduce the concept of "debt intolerance" and define it as the inability of emerging markets to manage levels of external debt that are manageable for advanced industrial countries. In their view, lower credit ratings are due to poor credibility (proxied by high levels of past inflation) and a history of repeated default.¹⁵ Finally, Mendoza (2003) emphasizes the role of limited and volatile tax bases.16

Whatever the cause (and irrespective of whether there is a solution or not), there is a consensus that under the current rules of the game, EMs can safely manage fairly low levels of external debt. While IMF (2002) sets the threshold for a safe level of debt at around 40 percent of GDP, Reinhart, Rogoff and Savastano (2003) are even more pessimistic and argue that some countries may not safely manage levels of external debt that are above 15 percent of GDP.

Non-renewable resources may affect the outcome of standard sustainability calculations. Chalk and Hemming (2000) show that, in presence of non-renewable

¹⁵ For a discussion of the differences between currency mismatches, debt intolerance, and original sin, see Eichengreen, Hausmann and Panizza (2003b).

¹⁶ Tax-to-GDP ratios in industrial countries average above 40 percent and 30 percent in developing countries. At the same time, volatility of tax revenues (measured as the coefficient of variation) tends to be twice as large in developing countries.

resources, sustainability would require equalizing a country's net worth (including the value of the nonrenewable resources) to the net present value of primary non-resource deficits. This highlights the fact that increasing resource exploitation to pay debt would not affect sustainability. There are, however, at least two problems with this Ricardian Equivalence of natural resources. The first was highlighted in our discussion of Buiter's indicator of fiscal sustainability, i.e., computing a country's net worth is a very difficult exercise. Second, it assumes that "oil in the ground" has the same return as the various financial assets and liabilities of the government.

Concessional debt also plays an important role in determining sustainability. In adopting any of the steady-state approaches described above it is important to recognize that as developing countries grow and become richer, the share of their concessional debt is bound to decrease and this will increase the cost of financing total debt. This increase in the cost of debt should be taken into account in estimating the steady state real interest rate paid by government debt.¹⁷

3.1 Other Issues

In standard IMF-led stabilization programs, countries are often asked to commit to achieving a given target in terms of primary deficit with the implicit objective of stabilizing or reducing its debt ratios.¹⁸ This approach has come under criticism for not taking into account two factors: (i) not all debt has the same level of risk; and (ii) deficits incurred to finance public investment should be treated differently from deficits incurred to finance current expenditure.

Equation (6) shows that the stock of public sector debt cannot be considered a monolithic entity. We have already discussed the fact that different types of debt have different implications for the volatility of the public sector deficit and, hence, for sustainability. Local currency fixed-term debt is clearly the safest form of debt because the cost of servicing this debt is by and large predetermined. Foreign currency official debt is also relatively safe. While it is subject to exchange rate risk, the interest rates charged by

¹⁷ For a discussion of the relationship between concessional debt, debt relief and fiscal sustainability, see Edwards (2002).

¹⁸ For a survey, see Chalk and Hemming (2000).

official creditors tend to be stable and the flow of financing is either cyclical or slightly counter-cyclical. Local currency floating rate debt is subject to interest rate volatility and the cost of servicing this kind of debt may increase substantially during periods of financial turmoil. Foreign currency non-official debt is the riskiest. The cost of servicing this debt is subject to the volatility of both the exchange rate and of sovereign risk (the latter applies only if the debt needs to be rolled over). Furthermore, sudden stops in capital flows may impede the ability of emerging market governments to roll over the debt leading to both a liquidity and solvency crisis.

On the basis of these considerations, it has been argued that debt sustainability exercises should be performed by making use of weighed debt-to-GDP ratios. Where, akin to the Basel principles for evaluating banks' balance sheets, different types of debt should be weighed according to their risk.

The second issue relates to the way in which investment expenditures should be recorded in government accounts. The debate is motivated by the concern that, according to current practice, public sector adjustment strategies bundle together current expenditure and public investment. The Rio Group (a permanent mechanism of political consultations and interaction between 19 Latin American countries) put forward a proposal aimed at excluding investment expenditure from fiscal deficit targets. The main argument in favor of this proposal is that as current expenditure tends to be difficult to adjust (because it is mostly composed of wages and entitlement programs) investment is the typical adjustment variable when the deficit exceeds the target. The proposal argues that the inclusion of investment expenditures in the target budget balance may be problematic because it basically considers every increase in debt as a reduction in government wealth, implicitly assigning no value to investment expenditure as an addition to net wealth. The Rio Group, instead, would favor the adoption of sustainability indicators similar to the one proposed by Buiter (equation 2).

We have already mentioned that a potential problem with targets based on net worth is that net worth is very difficult to compute. Furthermore, this issue is linked to the role that the public sector in an EM should play in terms of its investment strategy. EMs typically face periods of financial constraints where capital markets close, and are subject to credibility problems regarding their ability to remain sustainable in times of distress.

Crises occur in a context in which expenditure remains inflexible, particularly given its high wage content, and there are a myriad of unsatisfied social demands facing EMs. Under this scenario, it may be beneficial for the government to remain involved only in cofinancing investment activities that would otherwise not be carried out completely by the private sector, and that are deemed to be socially profitable (instead of assuming responsibility for full financing of investment projects). This, of course, does not mean that governments should not invest, but that they should do so selectively, complementing private sector participation. If this principle is accepted, it follows that the share of a project that is financed by the government is essentially a subsidy, and as such should be computed "above the line." This does not imply that public investment will always result in a higher fiscal deficit, because the possibly higher fiscal revenue that such investment might entail should be taken into account (and which, again, should be computed "above the line"). Of course, to the extent that governments are not following this rule, there is still an issue regarding the need for accounting public investment as an addition to net wealth, but then again, there are several factors that make valuation of this investment quite difficult, particularly because this capital is not easily marketable (it is basically a non-tradable good). Also, keeping a separate budget for current expenditures and investment expenditures may make it much harder to ensure sustainability.

4. EVALUATING SUSTAINABILITY UNDER UNCERTAINTY

Standard sustainability analysis starts by picking values for steady state growth and the real interest rate, and then uses these values to compute the level of primary surplus that is consistent with debt stabilization. The previous section highlighted that evaluating sustainability in emerging markets and developing countries is more difficult than evaluating sustainability in industrial countries because emerging market countries are characterized by higher volatility in both revenue and expenditure, and, hence, the various implications based on steady state values for growth and the real interest rate (where volatility is not an issue) may not make much sense.

In this vein, a policy paper published by the International Monetary Fund (IMF, 2003b) describes several methodologies to stress test standard sustainability analysis with shocks to the main macroeconomic variables (this could be done by observing the past

volatility of the main variables or by using stochastic simulations) and explicitly including contingent liabilities. These simulations can then be used to build confidence intervals around the projected evolution of debt over GDP.

To address the issue of volatility, some authors are now developing probabilistic models of sustainability that specifically take into account volatility in macroeconomic variables. Barnhill and Kopits (2003) develop models based on the concept of value-at-risk and apply it to the case of Ecuador. Hausmann (2003) also uses the value-at-risk concept and applies it in a cross-country context. Croce and Juan-Ramon (2003) develop a stochastic model aimed at deriving a fiscal policy rule that could be observable by external analysts and indicate whether a country is adopting a sustainable policy stance. Since surveying all these models would require a considerable amount of space and technicality, in our analysis, we will focus on the probabilistic model developed for the Inter-American Development Bank by Mendoza and Oviedo (2003). This is the model that we will use as a benchmark for our analysis, focusing on revenue volatility and expenditure inflexibility. Section 4.1 describes the main characteristics of the model. We then illustrate its application for the case of Egypt in section 5.1.

4.1 The Mendoza-Oviedo Model

The guiding principle of the Mendoza-Oviedo (MO) model is that of "credible payment commitment" (CPC). According to their definition, a commitment to repay is credible only if the government is able (not necessarily willing) to repay its debt in every state of nature.¹⁹ This implies that the government cannot accumulate more debt than the level it could service if it were to enter a fiscal crisis, defined as the case in which the primary balance remains forever at its lowest possible value. Were the actual level of debt to remain higher than the threshold determined by the CPC, then the government would be facing a positive probability of default on its debt, something that a risk-averse lender would not allow to happen.²⁰

¹⁹ Including the case in which government revenues (expressed as a share of GDP) remain indefinitely at their minimum.

²⁰ For more on this, including a discussion on outcomes using optimal contracts, see Mendoza (2003).

With these considerations in mind, Mendoza and Oviedo (2003) develop a fullblown dynamic stochastic general equilibrium model where the path of government revenues is endogenously determined by the behavior of utility-maximizing individuals and profit-maximizing firms, in a context where both tradable and non-tradable goods are produced. In their model, there is a mismatch in the government's balance sheet because the government debt is mostly denominated in tradables and tax revenues are mostly denominated in non-tradables. They also assume that volatility in government revenues can be traced back to volatility in fundamentals such as the terms of trade, foreign interest rates, or productivity. Since discussing the full specification of this model would require considerable technical detail, we follow a simplified version of Mendoza and Oviedo (2003) which transmits the essence of its more complex formulation. This simplified version is the one we use to conduct an application to Egyptian data in section 5.1.

We make the following three assumptions: (i) the path of government revenues is stochastic and exogenously determined; (ii) there is no currency mismatch (government revenues and debt are denominated in the same currency); and (iii) both the interest rate (r) and long-run GDP growth (g) are known with certainty.

These assumptions lead to a simple formulation of the credible repayment commitment where the threshold value for the debt-to-GDP ratio satisfies the following condition:²¹

$$d \le d^* \equiv (t^{\min} - e^{\min}) \frac{1+g}{r-g}.$$
 (7)

Here d^* represents the threshold value for the debt to GDP ratio, t^{\min} is the lowest possible realization of government revenues over GDP, and e^{\min} is the minimum level of government expenditure-to-GDP ratio that can be sustained if the country were to enter a fiscal crisis in which tax revenue reaches (and stays at) t^{\min} and pushes d above d^* . Heuristically, e^{\min} measures a government's ability to reduce expenditure in the presence of a prolonged negative shock to revenue. Countries that can sustain larger adjustments

²¹ Equation (7) could also be written as $d \le d^* = \frac{(t^{\min} - e^{\min})}{r - g}$. See footnote 7 for a discussion.

will be able to sustain larger debt-to-GDP ratios relative to countries with rigid public expenditure.

In this version of the model, the government has a constant desired level of primary expenditure, e. For each period (year), it observes the realization of revenue and finances any gap between revenue and total expenditure (including interest payments) with new debt, as long as the resulting debt does not hit the debt threshold d*. Otherwise, it needs to adjust expenditure in order to meet the debt threshold restriction. Thus, as long as debt is lower than the threshold, the debt-to-GDP ratio evolves according to the following formula:

$$d_{t} = d_{t-1} \frac{(1+r)}{(1+g)} + e_{t} - t_{t}$$
(8)

Given an initial debt level, and a sequence of revenue realizations based on the stochastic characteristics of the revenue process (mean, standard deviation, and persistence), the model generates a set of relevant results, despite its simplicity. Not only does it determine a threshold debt level, but it also produces estimates for the number of periods it will take to reach the debt threshold, as well as a probability distribution of debt n periods ahead. This probability distribution can then be used to calculate the probability of reaching the debt threshold.

An important difference between the probabilistic approach to sustainability of Mendoza and Oviedo and the traditional long-run approach is that the traditional approach defines a policy target (expressed as the primary-balance-to-GDP ratio) aimed at stabilizing the current debt-to-GDP ratio (which is assumed to be the steady state level of debt to GDP). In contrast, in the probabilistic model only the maximum level of debt to GDP is defined, but this level is not the equilibrium that will necessarily be observed and is clearly not the optimal level of debt. The task of the government is to strengthen fundamentals so that the probability of reaching the maximum level of government debt remains low.

An implication of the probabilistic model is that, for any given average revenue-to-GDP ratio, governments that have a less volatile revenue base (for instance, governments

that depend less on natural resources) will have higher values of $t^{\min 22}$ and hence they will be able to sustain higher levels of debt. Furthermore, what really matters is not the actual value of expenditure adjustment that a country can announce, but the value of e^{\min} that can be *credibly* announced. Countries that can commit to large adjustment in expenditure, can sustain higher debt-to-GDP ratios and may never be asked to act on these commitments.²³

Mendoza (2003) uses equation (7) to compute sustainable debt-to-GDP ratios under different assumptions for the volatility of revenues, the difference between GDP growth and the real interest rate, and the ability to adjust expenditure. He shows that the results are very sensitive to the choice of this last parameter. In particular, he finds that emerging market countries that cannot adjust expenditure by more than 1 percent of GDP will not be able to sustain positive debt. EMs with a larger capacity to adjust²⁴ may be able to sustain debt-to-GDP ratios that range between 40 and 150 percent. When he calibrates the results to the "average" emerging market countries²⁵ and 30 percent for high-risk emerging market countries. Comparable estimations calibrated for industrial countries yield a sustainable debt-to-GDP ratio that can reach up to 350 percent with average values of 85 percent.

Notice that these large differences in sustainability were obtained by just assuming differences in the volatility of revenues and the capacity to adjust primary expenditure. Hausmann (2003) suggests that valuation effects brought about by liability dollarization and original sin are likely to greatly amplify these differences. In fact, liability dollarization will affect the difference between real interest rate and GDP growth by making a given country riskier and hence play a role in determining which countries lie in the high-risk group.

²² Where, for example, t^{\min} is assumed to be equal to the mean minus two standard deviations.

²³ Countries could also commit to adjusting their tax rate and, by increasing t^{\min} , obtain a similar result.

²⁴ Defined as countries that can adjust the expenditure-to-GDP ratio by at least 8 percent.

²⁵ Low-risk is defined as having a small difference between real interest rate and GDP growth.

5. THE CASE OF EGYPT

This section surveys the main issues related to fiscal sustainability in Egypt. Figure 1 describes the evolution of the public deficit (expressed as share of GDP) over the period 1990-2004.²⁶ The trends are somewhat difficult to interpret because of a revision in the way in which the government reports the budget. For this reason, the figure reports the historical series (where the deficit is measured with the old reporting system) plus two series that start in 1998 and measure the deficit with the new reporting system. The first of the new series measures the deficit of the narrowly-defined budget sector and the second focuses on the total budget sector that includes the National Investment Bank (the institution that funds public projects), the General Agencies for Supply of Commodities (which administers the main subsidies), and the Social Insurance Fund. As the Social Insurance Fund runs a surplus (approximately 5 percent of GDP), the latter definition is the one that yields the lower deficit.²⁷



²⁶ The data for the period 1990-2002 are from the Central Bank of Egypt. The data for 2003 and 2004 are our own projections based on information obtained from the Economist Intelligence Unit.

²⁷ The government presents three sets of fiscal accounts. Besides the two described in Figure 1, it also reports a set of fiscal accounts that includes the activities of the National Investment Bank (NIB) and the General Agencies for Supply of Commodities (GASC), but not the Social Insurance Fund. As NIB and GASC have large losses, this third definition is the one that yields higher deficits. Part of the confusion in reading the budget numbers arises from the fact that the budget presented to the parliament does not use this three-tier methodology. Also, peculiar budget institutions always lead to large amendments and produce a final budget that is very different from the one that is initially proposed (see Panizza, 2002).

Whichever definition of the deficit one focuses on, Figure 1 clearly shows an inversion of the trend towards fiscal adjustment that characterized the early 1990s.²⁸ The deficit as a share of GDP bottomed out in 1997/98 and then started growing at a fast rate with the last figures placing the deficit well above 5 percent of GDP (2.5 percent of GDP after including the activities of the Social Insurance Fund). Furthermore, the deficit is projected to increase to 8.5 percent in 2003/04 (6 percent of GDP) after including the Social Insurance Fund.

This deterioration is partly due to the drop in GDP growth (Figure 2) caused by the large external and domestic shocks that have impacted the Egyptian economy in recent years. The Asian crisis of 1997 led to a reduction in portfolio flows and foreign direct investment and to a deterioration of terms of trade that affected Egypt's external account both directly (Egypt is a net oil exporter) and indirectly (through the remittances of Egypt's expatriates who work in Gulf countries).²⁹





1986/87 1987/88 1988/89 1989/90 1990/91 1991/92 1992/93 1993/94 1994/95 1995/96 1996/97 1997/98 1998/99 1999/00 2000/01 2001/02 2002/03 Source: CBE and authors' calculations based on EIU data.

²⁸ The fiscal consolidation of the late 1980s and early 1990s was helped by debt rescheduling and large international packages. A first major rescheduling took place in 1987 and a second one, mostly a reward for Egypt's role in the first Gulf War, occurred in 1991.

²⁹ There are about 1.9 million expatriates (more than 10 percent of total employment).

The Luxor incident of November 1997 and the events of September 11, 2001 also led to large drops in tourism revenues.³⁰

The large balance of payments deficit of 1998 was a shock for the Central Bank, which responded to this new situation in a rather incoherent manner. Monetary policy was tightened, then relaxed, then tightened again, and the exchange rate parity was defended with market-unfriendly policies that generated a black market for foreign currency that still exists. Some observers have suggested that policy uncertainty played a key role in amplifying the effect of the external shocks (Panizza, 2002).

It should be pointed out that, while the economic downturn of the last few years played an important role in the deterioration of Egypt's fiscal accounts, Egypt's budgetary problems are not purely cyclical. They have a structural component that needs to be addressed by a change in policy stance. This can be seen by observing the main components of Egypt's budget (Table 1). The upper part of the table focuses on the budget sector and the bottom part on the total budget sector; all figures are expressed as shares of GDP.

When we focus on the budget sector, we find an increase in current expenditure that is mostly due to a higher wage and interest rate bill and a decrease in capital expenditure. As these two trends compensate each other, total expenditure remained unchanged during the period under observation. This suggests that the deficit was due to lower revenues rather than higher expenditure. In fact, total revenues dropped from 24 percent of GDP in 1998/99 to 20.6 percent of GDP in 2001/02.

³⁰ In 1988, tourism surpassed oil and became the main source of hard currency. The tourism sector currently employs 2.2 million people and, according to official data, is responsible for 5 percent of GDP. Because of limited reserves, oil revenues are likely to further decrease and Egypt is estimated to become a net importer within the next 2 decades (EIU).

	Budget Sector					
	1998/99	1999/00	2000/01	2001/02	2002/03*	2002/04*
Total Expenditure	26.90	26.17	26.46	26.45	26.94	27.59
Current Expenditure	20.57	20.61	22.25	22.35	22.26	23.21
Wages	6.58	6.55	6.94	7.38	7.30	7.44
Interests	5.52	5.49	5.75	5.99	6.09	6.57
Local	4.73	4.96	5.25	5.38	5.52	5.88
Foreign	0.78	0.53	0.50	0.61	0.58	0.69
Capital Expenditure	5.90	4.94	4.16	3.99	4.37	4.23
Total Revenues	23.90	22.27	20.96	20.65	19.61	18.93
Deficit	3.00	3.90	5.50	5.80	7.33	8.67
Primary Deficit	-2.52	-1.59	-0.25	-0.19	1.23	2.10
	Budget Sector + NIB+GASC+SIF					
	1998/99	1999/00	2000/01	2001/02	2002/03*	2002/04*
Total Expenditure	30.95	30.08	30.02	29.71	30.26	31.00
Current Expenditure	22.65	22.52	24.77	23.84	23.75	24.76
Wages	6.65	6.62	7.01	7.45	7.37	7.51
Interests	4.88	4.82	5.18	5.32	5.41	5.84
Local	4.10	4.29	4.68	4.71	4.83	5.15
Foreign	0.78	0.53	0.50	0.61	0.58	0.69
Capital Expenditure	5.90	4.94	4.16	3.99	4.37	4.23
Total Revenues	30.87	28.85	27.81	27.20	25.84	24.93
Deficit	0.07	1.23	2.21	2.51	4.42	6.06
Primary Deficit	-4.81	-3.59	-2.98	-2.82	-0.99	0.22

 Table 1: Egypt's Main Budget Items (as a share of GDP)

Source: Authors' calculations based on CBE data and (*) projections based on data from the Economist Intelligence Unit.

While the external situation is now improving, the most recent budgets show no sign of improvement in fiscal accounts and the budget for 2003/2004 does not seem to correct the negative trend described above. In particular, a small drop in public investment will not be enough to compensate the drop in revenues and an increase in current spending (the main components are wages and pensions). Because of the deterioration of the fiscal accounts, Standard and Poor's downgraded Egypt's sovereign foreign currency debt from BBB- (investment grade) to BB+ (speculative) with a negative outlook.

Figure 3 describes the behavior of public debt expressed as a share of GDP. It separates total debt into public domestic and public external debt. The data reported on the figure (like all debt data used in this paper) refer to net debt. If we were to use gross debt,

we would obtain much higher debt to GDP ratios.³¹ Alba, Al Shawarby and Iqbal (Forthcoming) discuss the difference between net and gross debt. The figure shows that there is a tendency toward greater reliance on domestic debt. The share of domestic debt over total debt climbed from 40 percent in the early 1990s to about 65 percent in 2000/2001. The Egyptian government has also been successful in lengthening the maturity of its debt. In 1995, it started placing 5-year domestic currency bonds and currently issues bonds with maturities of up to 7 years. Most of this debt, however, is held by domestic banks. If these bonds are financed with demand deposits, the fact that the government is able to place long-term bonds does not reduce the aggregate maturity mismatch, but it only transfers it to the banks' balance sheets.



Figure 3: Public Debt Over GDP

³¹ Net debt is difficult to estimate. While we use official figures form the Central Bank of Egypt, IMF estimates (release of Article IV consultation for 2000) suggest that netting out government deposits with the Central Bank and a blocked account created at the time of the Paris Club rescheduling would reduce the stock of domestic debt by approximately 5.5 percentage points. This would bring debt to GDP in 2000/2001 to approximately 66 percent (versus 71.5 percent reported in Figure 3). While different methods of calculation affect the level of debt, they should not affect the trend described in Figure 3.

Figure 4 looks at the composition of external debt. Most public external debt is long and medium-term, but in the last 6 years the share of short-term debt almost doubled (from 4 to 7.5 percent). The figure also shows that official creditors (mostly bilateral) hold about 90 percent of medium and long-term external debt (approximately 70 percent of long-term debt is granted on concessional terms). However, the share of public external debt held by private creditors has increased substantially in the last few years, from 10 percent to approximately 14.5 percent (this calculation assumes that all short-term debt is held by private creditors).



The fact that only a small share of external debt is short-term and held by private creditors suggests that Egypt's debt structure is fairly safe (from the borrower's point of view). Sudden stops in capital flows or large jumps in country risk due to international contagion are unlikely to have a large negative effect on Egypt's external balance and fiscal accounts. There is also little reason to be worried about a sudden stop in international aid. While net official development assistance decreased from \$2.2 billion in 1996 to \$1.3 billion in 2000, the World Bank and the United States government recently pledged large grants and loans.

The events of 1998, however, show that an unfavorable external climate may cause a sudden drop in foreign revenues and have an effect that is similar to that of a sudden stop in capital flows. Figure 5 shows that during 1997-1999, three components of the balance of payment (oil, tourism revenues, and portfolio flows) were responsible for a drop in foreign currency revenues equivalent to 6 percent of GDP. Such a sudden stop in foreign revenue may lead to a currency depreciation that may have serious valuation effects. According to the Central Bank of Egypt, the jump in external debt observed in 2001/2002 (from 28 to 32 percent of GDP) was mostly due to the depreciation of the Egyptian pound. While the depreciation of the pound is expected to further increase external debt to well above 40 percent of GDP, one should remember that the government has large foreign currency deposits with the Central Bank and with the banking system that partly offset the negative effect of a currency depreciation. In fact, some analysts think that the currency depreciation had either a neutral or even a positive effect on the overall government balance.





5.1 Sustainability Analysis

Motivated by Egypt's recent developments, we now focus on three fiscal sustainability exercises: (i) a standard sustainability analysis; (ii) the Mendoza-Oviedo (2003) probabilistic approach; and (iii) the Calvo-Izquierdo-Talvi (2002) sudden stop approach.

These three approaches are useful in assessing different sustainability issues. The traditional approach, which equates the current debt to GDP ratio to the steady state debt level, is useful to calculate the primary surplus that is consistent with that debt at alternative interest rates and growth rates. The second approach incorporates revenue volatility and expenditure flexibility to the previous case. Finally, the third approach is concerned with the effects of an external shock that leads to adjustment in the current account balance and real exchange rate depreciation.

Standard Approach to Sustainability

The standard approach requires assumptions on initial debt, steady-state GDP growth and steady-state real interest rate. For our baseline calculation, we assume that the starting level of debt is 71.5 percent of GDP, i.e. the value for net public debt prevailing in 2002, and we let the steady state growth rate range between 2 and 5 percent and the real interest rate range between 6 and 12 percent. Figure 6 shows that under the most favorable conditions (6 percent real interest rate and 5 percent growth rate of the economy), the Egyptian government could stabilize the debt with a primary surplus equivalent to 0.7 percent of GDP. If we move away from this scenario, however, we find that required primary surpluses go well above 3 percent of GDP. In particular, if we assume that steady-state long-run growth is 4 percent and the real interest rate is 10 percent, we find a required primary surplus of 4.1 percent of GDP. This is a much higher figure than the surpluses that the government has been able to run in the past few years.



Figure 6: Primary Surplus Required to Stabilize Debt at 71 percent of GDP

Probabilistic Approach to Sustainability

The Mendoza-Oviedo probabilistic approach is useful in illustrating how revenue volatility can be incorporated in a sustainability analysis. Given methodological changes in fiscal data reporting and the lack of consistent time series, we need to take some shortcuts. We first calculate volatility based on a measure of the cyclical component of government revenue (obtained by taking a Hodrick-Prescott filter to the log of revenue) in real terms for the period 1981-2002.³² Next, we approximate cyclical component behavior to a first order autoregressive (AR1) process, and take the standard deviation of this process as our measure of volatility.³³ This yields a value of 2.5 percent.

We then construct measures of average revenue and average non-interest expenditure for a wider measure of the public sector, which includes the National Investment Bank and GASC, over the period 1999-2002.³⁴ These measures yield 24.9

³² We use data on budget sector revenues. We could not incorporate a wider definition of the public sector (including the National Investment Bank and GASC) in our estimation of revenue volatility given the lack of sufficiently long time series, particularly after the changes in definition that took place in 1998.

³³ This measure of volatility is expressed in percentage terms relative to the Hodrick-Prescott trend. We also use the estimated autocorrelation coefficient of the AR1 process to generate a Markov chain, defining transition probabilities for the revenue process. The latter are used later to simulate revenue paths.

³⁴ These were the only years for which information is available under the new reporting system of the CBE.

and 22.3 percent of GDP, respectively. Next, we estimate the effective real interest rate paid by the central government based on interest payments and prevailing debt stocks at the beginning of the period.³⁵ In order to avoid the effects of substantial recent devaluation, we compute average interest payments for 2001.³⁶ This estimate includes the effect of real exchange rate depreciation that took place in 2001 (given that the nominal exchange rate depreciated at a faster rate than the increase in domestic prices), thus raising the value of the real interest rate paid on dollar debt. With this caveat in mind, we estimate the average real interest rate paid on government debt to be 10.6 percent. Given that we measured volatility for the period 1981-2002, we use the same time span to estimate the average growth rate of the economy,³⁷ yielding a value of 4.1 percent. Finally, we take net public debt prevailing in 2002 as our initial debt level.³⁸ This figure is equivalent to 71.5 percent of GDP.

At this stage, we need to make further assumptions about minimum revenue, as well as the flexibility of public expenditure. We make the following two assumptions: (i) the minimum revenue level lies two standard deviations below the mean (this is equivalent to a probability of 2.5 percent); given our measures for volatility and mean revenue, this yields a value of 20 percent of GDP; (ii) the government can reduce non-interest expenditure by 30 percent in the event of a fiscal crisis.

The assumptions on revenues volatility and ability to adjust non-interest expenditure are particularly important and merit further discussion. In particular, we assumed an ability to adjust expenditure that is much larger than the one that is suggested by the cross-country evidence. Figure 7 is helpful to depict debt threshold sensitivity to revenue volatility and expenditure adjustment. Taking interest and growth rates as given,

 $r = \frac{(1+r^*)(1+\varepsilon)}{1+\pi}\alpha + \frac{1+r^d}{1+\pi}(1-\alpha) - 1$, where r* is the interest rate on dollar debt, ε is the devaluation rate, π is the inflation rate, r^d is the interest rate on domestic currency debt, and α is the share

³⁵ Data on interest payments and net debt is obtained from CBE sources.

³⁶ The real interest rate facing the government (r) is defined as follows:

of foreign debt in total debt.

³⁷ GDP data was obtained from IMF (2003d).

³⁸ CBE data.

different debt thresholds are calculated for various values of volatility and adjustment levels. For example, using as a benchmark our estimate for revenue volatility (2.5 percent of the mean) and our assumption on ability for expenditure adjustment (30 percent), the debt threshold yields 77 percent of GDP. Reducing volatility to 1.7 percent would allow for higher debt levels (100 percent of GDP).

On the other hand, if we were to relax our highly unrealistic assumption that the government can adjust expenditure by 30 percent and replace it with the still generous assumption that government can adjust expenditure by 20 percent (an even more realistic assumption would put the maximum adjustment at around 15 percent), we would obtain a debt threshold close to 45 percent of GDP. This would imply that current debt is already above the threshold that guarantees debt repayment under all states of nature and that fiscal adjustment is necessary.





While we were very lenient in assuming a large capacity to adjust, we were stricter in calculating the interest rate paid on government debt. If we were to assume that most of the adjustment in the exchange rate has already taken place, the average interest rate on

government debt will be lower than the one used in our calculations, thus yielding a higher debt threshold.

This simple exercise points out that Egypt's fiscal position is far from being lenient. The current level of debt appears to be sustainable (in the probabilistic definition) only if we are ready to assume unrealistically high flexibility in government expenditure. The definition of sustainability is very conservative because it insures the public sector against default even under the worst scenario (in which revenue remains at its minimum forever) and it may be argued that this is a very unlikely event. The threshold level obtained with the probabilistic sustainability analysis is a good benchmark for comparison against actual debt levels, particularly so for developing countries where uncertainty about the future path of revenue and credibility about debt repayment is always an issue. Furthermore, this exercise has the advantage of highlighting that macroeconomic policies that reduce revenue volatility and increase flexibility in public spending are two key determinants of access to credit markets.³⁹

We now turn to model simulation. We assume a given level of initial debt and perform 1000 replications of the model to estimate the average number of periods it takes to hit the debt threshold, given all above-stated assumptions. Figure 8 shows the results for various initial debt levels. If we take as initial level of debt, the one prevailing in 2002 (71.5 percent) we find that it would take on average 7 periods to reach the debt threshold. However, given the high volatility of government revenue, the standard deviation of the number of periods in which debt can hit the threshold is also high (implying that the threshold could be reached sooner). While this is a simple exercise subject to further fine-tuning, it suggests that Egypt may have to enter a fiscal adjustment phase relatively soon.

³⁹ So far, we have treated debt thresholds as triggers of an adjustment phase in government expenditure. Alternatively, if authorities could not come up with the necessary adjustment assumed initially to be feasible by lenders, it would be equivalent to a default situation. Thus, hitting the threshold could have an interpretation of hitting default.



This simple model also allows computing the relative frequency distribution of government debt *n* periods ahead. This distribution contains all the information needed to compute the probability of reaching the debt threshold. Figure 9 displays debt distributions 3, 6, and 9 periods (years) ahead. The first graph indicates that, under the favorable assumptions described above, the probability of entering an adjustment phase within three periods is rather low (3.8 percent). This changes substantially if we focus on a six-period horizon. In this case, even with our favorable assumptions, the probability of entering the adjustment phase jumps to 53.5 percent. The probability of entering the adjustment phase reaches 87.9 percent if we look at a nine-period horizon.

Figure 9 illustrates one of the key advantages of the probabilistic approach which, by allowing estimation of the future behavior of debt, can highlight possible negative outcomes with enough anticipation, giving authorities a chance to correct fiscal policy well before hitting debt thresholds.



Figure 9: Public Debt Distribution





Sudden Stop Approach to Sustainability

We now focus on the fiscal distress that is typically generated by shocks to financing of the current account, or sudden stops in capital flows. As stated in Calvo, Izquierdo and Talvi (2003), this type of shock has been quite relevant for emerging markets, particularly so following the Russian crisis of 1998. These events can be interpreted as shocks to credit. A fall in the financing of the current account deficit implies a forced adjustment in the absorption of tradable goods. To the extent that consumption of non-tradable goods is a complement in consumption of tradable goods, a fall in the latter will imply a fall in the former, leading to a decrease in non-tradable prices. Since for a small open economy tradable prices are taken as a given, this implies that the real exchange rate will have to adjust. Adjustment in the real exchange rate will generate valuation effects on the debt-to-GDP ratio, which, in turn, affect fiscal sustainability.

The case of Egypt is different from the typical EM in that its current account deficit is small. For example, according to IMF (2003d), it reached only 0.01 percent of GDP (or 0.04 percent of imports) in 2002, a much smaller figure than pre-crisis levels in countries like Argentina or Ecuador, where current account deficits exceeded 4 percent of GDP and 9 percent of GDP, respectively. Thus, the shock to financing of the current account deficit, from a capital flow perspective, would be very small for the case of Egypt. Yet, there are alternative sources of vulnerability that could work as sudden stops. If the government faced restrictions to increase its debt levels (as would be the case under the Mendoza-Oviedo framework), then shocks such as a fall in the price of oil or a decrease in remittances may generate effects similar to those of a sudden stop, given that there may be no additional sources of financing available to smooth out the shock.

In order to assess the significance of this type of shock to the Egyptian economy, we focus on a scenario where the price of oil goes back to 1999 levels, equivalent to a fall in real terms relative to current prices of 49 percent. Under the assumption that oil output remains constant, we compute this percentage price fall as the percentage fall in the value of oil exports. Thus, financing could fall by as much as US\$1 billion, equivalent to 5 percent of imports on impact.⁴⁰ Assuming that the latter represents the percentage fall in

⁴⁰ We take this as a proxy of the fall in absorption of tradable goods following the shock.

the absorption of tradable goods, that a similar percentage fall in the demand for nontradable goods would take place,⁴¹ and that the relative price elasticity of non-tradable goods has a value similar to that of other developing countries,⁴² this would yield an increase in the real exchange rate of about 11 percent.⁴³ This figure does not seem to be very high, at least compared to the effects that sudden stops have had in Latin America, where the real exchange rate skyrocketed in the aftermath of a sudden stop, particularly because of the large current account deficits that many of these countries were running before the crisis.

We focus next on the effect of real depreciation on fiscal sustainability through its impact on the debt-to-GDP ratio. A remarkable characteristic of the Egyptian economy is the currency composition of government debt. As of 2002, domestic currency debt was 15 percent higher than foreign currency debt, something quite uncommon for a developing economy, where, typically, an inflationary past has led to the acceptance of foreign currency debt only. On one hand, this fact could be interpreted as a healthy sign, indicating that expectations of inflationary repudiation are low. On the other hand, it may be reflecting that much of domestic public debt acceptance is due to financial repression and large government bond holdings by public banks.⁴⁴ Having said this, at least from the non-financial public sector perspective, a substantial share of government debt seems to be shielded from real exchange rate fluctuations.

We analyze this in more detail by taking a look at government mismatches in debt composition *vis-à-vis* output composition. It can be shown that real exchange rate fluctuations will have no valuation effects on the debt-to-GDP ratio as long as the ratio of debt in domestic currency (or in non-tradables, B) relative to debt in foreign currency (or in tradables, B*) is the same as the ratio of non-tradable output (Y) to tradable output (Y*), or equivalently:

⁴¹ Given that utility functions are assumed to be homothetic.

⁴² This value is taken to be 0.4.

⁴³ An increase is interpreted as an increase in the price of tradable goods *vis-à-vis* that of non-tradable goods. For a detailed description of this calculation, see Calvo, Izquierdo, and Talvi (2003).

⁴⁴ Data on government bond holdings held by public banks was not available, although commercial banks as a group (including public banks) hold a substantial amount of those assets, and public banks represent a relevant share in total commercial banks, at least in terms of their number of branches relative to total commercial bank branches.

$$\frac{B}{B^*} \frac{Y^*}{Y} = 1 \tag{9}$$

Compared with other developing countries, Egypt ranks very well in terms of its B/B* ratio, but it has average values in its Y/Y* ratio.⁴⁵ The measure $\frac{B}{B*}\frac{Y*}{Y}$ yields a value of 0.25, much larger than that of crisis countries like Argentina or Ecuador (where this ratio was close to zero) and closer to that of sudden stop survivors such as Chile, where this ratio yielded 0.48.

We put Egyptian data to the test by analyzing the effect on the required primary surplus (using standard sustainability analysis) of a real exchange rate (RER) depreciation triggered by the fall in the price of oil indicated above. The results are shown in Table 2.

	Case 1	Case 2
(B + B *) / (Y + Y *)	71.47	71.47
r	10.57	6.79
g	4.11	4.11
Observed Primary Surplus	0.19	0.19
Required Primary Surplus		
a. Base Exercise	4.43	1.84
b. Change in Relative Prices	4.56	1.89
c. b + Increase in Interest Rate of 100 bps	5.27	2.60
d. c + Decrease in GDP Growth of 1%	6.03	3.34
e.d + Contingent Liabilities (3.4% GDP)	6.33	3.50
	0.05	
(B / B *) / (Y / Y *)	0.25	0.25
Fall in Oil Exports/Imports of G & S	0.05	0.05

 Table 2: Egypt: Fiscal Sustainability After a Shock to the Current Account (Fall in Oil Prices, Data as of 2002)

We first use the same assumptions on interest and growth rates used in the Mendoza-Oviedo framework, and then conduct a second exercise with a different assumption on the real interest rate. As expected, the RER valuation effect has a very small impact on the required primary surplus (0.13 points of GDP). There are two reasons for this: (i) the

⁴⁵ Where Y* is proxied by exports. This result will change in a more thorough analysis of output composition, although, as shown in Calvo, Izquierdo and Talvi (2002), sustainability calculations may not differ substantially.

increase in the RER is small;⁴⁶ and (ii) debt mismatches are not very substantial. However, it is not uncommon for negative shocks to the current account to be accompanied by increases in interest rates and economic crises (with the consequent drop in GDP growth). An increase in interest rates of 100 basis points (1 percent) would increase the required primary surplus by 0.71 points of GDP, while a 1 percent drop in output growth would require an additional adjustment of 0.77 points of GDP.

Finally, we also explore the potential impact that contingent liabilities in the financial sector could have on the government. In many recent crises, governments have bailed out banks in trouble, particularly because of the emergence of non-performing foreign currency loans handed to non-tradable sectors that turn bankrupt after a substantial rise in the RER. In this respect, the Egyptian banking system does not seem to be particularly vulnerable to valuation effects in non-tradable sectors. Adding up private and public banks, credit in foreign currency to the private business sector represents slightly more than 20 percent of total credit to that sector.⁴⁷

Once we roughly split foreign currency private business sector credit into tradable (agriculture and industry) and non-tradable activities (trade and services), we found that only about 12 percent of total credit to the private business sector is allocated to non-tradable activities, representing about 7 percent of GDP. Assuming half of these loans turn bankrupt following a rise in the RER, and that the government incorporates this amount into government liabilities following a bailout, public sector debt would increase by 3.5 points of GDP. This increase in debt would, in turn, require an additional adjustment in the primary surplus of 0.3 points of GDP. As can be seen, none of these elements individually would add significantly to the adjustment requirement, but put *together*, they could introduce stress in the government fiscal position, particularly so once we consider that there is already a *relevant gap between the observed primary surplus and the required one, even in the absence of an external shock*.

⁴⁶ Where the rise, once again, is interpreted as a rise in the price of tradable goods relative to that of non-tradable goods.

⁴⁷ In turn, the private business sector gets the bulk of credit to the private sector, which represents about 86 percent of that total. Other claims of the banking sector include loans to the public sector, which are presumably allocated in domestic currency, thus leaving our non-tradable sector mismatch calculation unaffected.

As previously stated, the real interest rate used in the calculations (10.6 percent) is that obtained for 2001, when a rise in the RER of close to 10 percent substantially increased the real interest rate on dollar debt, and raised the effective real interest rate paid by the government. If one were to exclude this effect, assuming that in the future no additional RER adjustment would take place, the effective real interest rate would fall to 6.8 percent. The second column of Table 2 shows results for the same analysis carried out previously at the new interest rate. Again, although the effects are now obviously smaller, there is still a substantial gap between the observed primary surplus and the required surpluses both before and after the external shock.

6. CONCLUSION

In this paper, we survey the recent literature on fiscal sustainability with special focus on emerging market countries. We highlight that, because of greater uncertainty and high revenue volatility, standard sustainability analysis is not easily applicable to emerging market countries. With these considerations in mind, we describe in detail a model that aims at evaluating sustainability by using probabilistic methods. We conclude our discussion with an application to the case of Egypt.

Table 3 summarizes our main findings. On one hand, we find that Egypt's debt structure is rather safe and that the country is not likely to suffer from international contagion or from the disruptive sudden stops in capital flows that were at the basis of several financial crises in emerging market countries. Furthermore, our calculations suggest that even a worst case scenario – with a 50 percent drop in the price of oil, a jump in the steady state interest rate, a drop in steady state growth, and widespread bankruptcies – would add, at most, 2 points of GDP to the debt stabilizing primary surplus (not that this is an irrelevant matter, but the calculations were made under a very pessimistic scenario).

	-
Possible Source of	Discussion
Vulnerability	
Sudden Stops in Capital Flows	NO . Most of the external debt is long term and held by official creditors
Terms of trade shocks coupled with	UNLIKELY . It will have significant effect on required primary
large real depreciation	surplus only under extreme circumstances
Large structural Deficit	YES. Current primary surpluses are well below what is required to stabilize the debt to GDP ratio
High level of current debt	YES. Under current policies, Egypt may be close to hitting a
	level of debt that will require fiscal adjustment

 Table 3: Egypt Sources of Vulnerability

Furthermore, our calculation did not consider that: (i) Egypt's oil exports are broadly balanced by oil imports; (ii) in terms of the fiscal balance, a depreciation of the currency has uncertain effect because it affects both government expenditure and government revenue; and (iii) although tourism can be volatile, it has always recovered quickly after shocks. Therefore, we conclude that sudden stops should not be an important cause of concern in the case of Egypt.

On the other hand, even without shocks and under rather benign assumptions, fiscal sustainability in Egypt is a real concern. Observed primary surpluses are well below those that would guarantee debt stabilization, and simulations of the probabilistic model suggest that, under the current policies, Egypt is either already in a state of fiscal difficulty or is not far from reaching a debt threshold that would require a substantial fiscal adjustment.⁴⁸ It would therefore be advisable that the Egyptian government immediately start a fiscal adjustment program.

⁴⁸ One point that is worth making is that the government has a captive financing source from the public banks. A scenario of unsustainable fiscal policies would therefore more likely take the form of bank bailouts, rather than a refusal of these banks to lend to the government.

REFERENCES

- Alba, Pedro, Sherine Al Shawarby and Farrukh Iqbal. Forthcoming. "Debt Sustainability in Egypt." *ECES Working paper*.
- Barnhill, Theodore and George Kopits. 2003. "Assessing Fiscal Sustainability under Uncertainty." *IMF Working paper* 03/79.
- Blanchard, Olivier. 1990. "Suggestions for a New Set of Fiscal Indicators." OECD, Paris.
- Bohn, Henning. 1995. "The Sustainability of Budget Deficits in a Stochastic Economy." Journal of Money, Credit, and Banking 27(2): 252-272.
- Buiter, Willem. 1985. "Guide to Public Sector Debt and Deficits." *Economic Policy: A European Forum*, 1: 949-963.
- Calvo, Guillermo. 2003. "Explaining Sudden Stop, Growth Collapse, and BOP Crisis: The Case of Distortionary Output Taxes." *IMF Staff Papers*, 50, Special Issue.
- Calvo, Guillermo. 1998. "Capital Flows and Capital-Market Crises: The Simple Economics of Sudden Stops." Mimeo, University of Maryland.
- Calvo, Guillermo and Carmen Reinhart. 2002. "Fear of Floating." *Quarterly Journal of Economics* 117, 379-408.
- Calvo, Guillermo, Alejandro Izquierdo, and Ernesto Talvi. 2002. "Sudden Stops, the Real Exchange Rate, and Fiscal Sustainability: Argentina's Lessons." Inter.-American Development Bank, Research Department *Working Paper* 469.
- Calvo, Guillermo, Alejandro Izquierdo, and Luis F. Mejía. 2003. "On the Empirics of Sudden Stops." Mimeo, Inter-American Development Bank.
- Central Bank of Egypt. Annual Time Series data. Available at: www.cbe.org.eg.
- Chalk, Nigel and Richard Hemming. 2000. "Assessing Fiscal Sustainability in Theory and Practice." *IMF Working Paper* 00/81.
- Croce, Enzo and Hugo Juan-Ramon. 2003. "Assessing Sustainability: A Cross-Country Comparison." *IMF Working Paper* 03/145.
- Cuddington, John. 1996. "Analyzing the Sustainability of Fiscal Deficits in Developing Countries." Mimeo, Georgetown University and the World Bank.
- Economist. 2002. "Re-engineering the Euro." October 19.
- Economist Intelligence Unit (EIU). Egypt Country Report. Various issues.
- Edwards, Sebastian. 2002. "Debt Relief and Fiscal Sustainability." *NBER Working Paper* 8939.

- Eichengreen, Barry and Ricardo Hausmann. 1999. "Exchange Rates and Financial Fragility." In *New Challenges for Monetary Policy*. Kansas City, Missouri: Federal Reserve Bank of Kansas City, pp.329-368.
- Eichengreen, Barry, Ricardo Hausmann and Ugo Panizza. 2003a. "The Pain of Original Sin.". In *Debt Denomination and Financial Instability in Emerging-Market Economies*. Edited by Barry Eichengreen and Ricardo Hausmann. Chicago: University of Chicago Press.
- Eichengreen, Barry, Ricardo Hausmann and Ugo Panizza. 2003b. "Currency Mismatch, Debt Intolerance and Original Sin: Why they are Not the Same and why it Matters." *NBER Working paper* 10036.
- Galindo, Arturo, and Alejandro Izquierdo. 2003. "Terms of Trade Shocks, Sudden Stops, and Employment in Emerging Markets." Mimeo, Inter-American Development Bank.
- Hamilton, James and Marjorie Flavin. 1986. "On the Limitations of Government Borrowing: A Framework for Empirical Testing." *American Economic Review*, 76: 809-819.
- Hausmann, Ricardo, Ugo Panizza, and Ernesto Stein. 2001. Why Do Countries Float the Way They Float." *Journal of Development Economics* 66: 387-414.
- Hausmann, Ricardo, and Ugo Panizza. 2003. "The Determinants of Original Sin: An Empirical Investigation." *Journal of International Money and Finance* 22: 957-990
- Hausmann, Ricardo. 2003. "Good Debt Ratios, Bad Credit Ratings: The Role of Debt Structure." Mimeo, Harvard University.
- Inter-American Development Bank. 1995. Overcoming Volatility, Report on Economic and Social Progress in Latin America. Baltimore, Johns Hopkins University Press.
- International Monetary Fund. 2003a. "Debt Sustainability in Low Income Countries— Towards a Forward Looking Strategy." Policy paper prepared by the Policy Review and Development Department, May 23.
- International Monetary Fund. 2003b. "Sustainability Assessments—Review of Application and Methodological Refinements." Policy paper prepared by the Policy Review and Development Department, June 10.
- International Monetary Fund. 2003c. "Public debt in Emerging Markets: Is it too High?" *World Economic Outlook*, September 2003.

International Monetary Fund. 2003d. International Financial Statistics CD-ROM.

- International Monetary Fund. 2002. "Assessing Sustainability." Policy paper prepared by the Policy Review and Development Department, May 28.
- Mendoza, Enrique. 2003. "An Analytical Review of Public Debt Sustainability Analysis and its Implication for Emerging Markets." Mimeo, University of Maryland.
- Mendoza, Enrique and Pedro Marcelo Oviedo. 2003. "Public Debt Sustainability under Uncertainty." Mimeo, Inter-American Development Bank.
- Panizza, Ugo. 2002. "Macroeconomic Policies in Egypt: An Interpretation of the Past and Options for the Future." *Journal of Development and Economic Policies*, 5(1):5-37.
- Reinhart, Carmen, Kenneth Rogoff and Miguel Savastano. 2003. "Debt Intolerance." *Brookings Papers on Economic Activity* 1, pp.1-74.
- Wilcox, David. 1989. "The Sustainability of Government Deficit: Implications of the Present-Value Borrowing Constraint." *Journal of Money, Credit, and Banking* 21(3): 291-306.